THE COST OF MARKET POWER IN BANKING: SOCIAL WELFARE LOSS VS. COST INEFFICIENCY

Joaquín Maudos Juan Fernández de Guevara

FUNDACIÓN DE LAS CAJAS DE AHORROS DOCUMENTO DE TRABAJO Nº 248/2006 De conformidad con la base quinta de la convocatoria del Programa de Estímulo a la Investigación, este trabajo ha sido sometido a evaluación externa anónima de especialistas cualificados a fin de contrastar su nivel técnico.

ISBN: 84-89116-07-5

La serie **DOCUMENTOS DE TRABAJO** incluye avances y resultados de investigaciones dentro de los programas de la Fundación de las Cajas de Ahorros. Las opiniones son responsabilidad de los autores. The cost of market power in banking: social welfare loss vs. cost inefficiency

Joaquín Maudos (Ivie & Universitat de València) Juan Fernández de Guevara (Ivie)

Abstract

This paper analyses the relationship between market power in the loan and deposit markets and efficiency in the EU15 countries over 1993-2002. Results show the existence of a positive relationship between market power and cost X-efficiency, allowing rejection of the so-called quiet life hypothesis (Berger and Hannan, 1998). The social welfare loss attributable to market power in 2002 represented 0.54% of the GDP of the EU15. Results show that the welfare gains associated with a reduction of market power are greater than the loss of bank cost efficiency, showing the importance of economic policy measures aimed at removing the barriers to outside competition.

Key words: market power, welfare loss, X-inefficiency, banking JEL: D40, G21

Instituto Valenciano de Investigaciones Económicas (Ivie), c/ Guardia civil, 22, Esc. 2^a, 1^o, 46020 Valencia (SPAIN). Email: <u>joaquin.maudos@ivie.es</u> [corresponding author]; juan.fernandez@ivie.es

Universitat de València, Departamento de Análisis Económico, Edificio departamental oriental, Avda. de los Naranjos, s/n; 46022 Valencia (SPAIN).

^{*}Authors acknowledge the financial support of the Spanish Savings Banks Foundation (FUNCAS). The paper is developed within the framework of the research programs of the Ministry of Science and Technology-FEDER (SEC2002-03375 and SEJ2004-00110). Authors gratefully acknowledge comments of the participants of the *XXI Jornadas de Economía Industrial*, (Bilbao, 8-9 September 2005) and the *XIII Foro de Finanzas* (CEMFI, Bank of Spain, Madrid, 17-18 November 2005).

1. Introduction

Economic theory emphasizes the gains that perfectly competitive markets represent over those in which market power exists, insofar as the existence of market power implies a net loss of social welfare. In the case of the banking sector, the analysis of market power is especially important because it translates into a higher cost of financial intermediation, a lower volume of savings and investment, and therefore lowers economic growth. However, often some level of market power is accepted because it is assumed that higher profits would reduce the risk and enhance the stability of the banking system.

The economic authorities have always been aware of the importance of reducing the levels of market power in banking markets so that they can be as competitive as possible. Thus, since the mid-1980s, both national and European authorities have adopted measures tending to the liberalization of banking markets with initiatives like the freedom to branch expansion throughout the country, the liberalization of interest rates, the opening of the sector to foreign competition, the elimination of compulsory investment coefficients, etc. More recently, the Second Banking Directive (implemented between 1991 and 1994 by the different European countries), the creation of the European Monetary Union in 1999, and the approval and implementation of the Financial Services Action Plan by the European Commission between 1999 and 2004 gave new impulse to the creation of a single European financial services market.

In the academic sphere there has also been great interest in the measurement of the degree of competition in banking markets. Thus, in recent years there have appeared a substantial number of studies that use different indicators of competition (Lerner index, Panzar and Rosse's test, Bresnanhan's mark-up test, conjectural variation parameter) with empirical applications whose purpose is to analyze competitive rivalry in banking markets¹.

In the specific case of European banking sectors, the results of the studies carried out show the existence of market power, not being possible to reject the situation of monopolistic competition (Bikker and Haaf, 2002; De Bandt and Davis, 2000; Fernández de Guevara *et al.*, 2006). In general, the studies referring to the European banking system have focused on measuring the degree of competition without analyzing the factors that may explain the existence of market power. Only Fernández de Guevara *et al.* (2005) try to identify the factors that explain this non-competitive behavior, using as an empirical reference the principal European banking sectors. Also,

¹ See a recent survey of this issue in Berger *et al.* (2003).

few studies have quantified the capacity of banking firms to set prices above the marginal costs in different banking products. As Fernández de Guevara *et al.* (2005) show using aggregate information on interest rates, the degree of competition varies depending on the banking product considered (consumer loans, mortgage loans, deposits, etc).

An important question that has not received attention in the European case is the analysis of the consequences of market power for the efficient management of banks. In markets where the pressure of competition is low there may be an incentive for managers not to concern themselves with reducing inefficiency. The rationality of this behavior can be justified basically for the following reasons. First, the managers may have fewer incentives to manage the firm efficiently because the capacity to establish a price above marginal cost generates sufficient profits to justify their management. Second, thanks to market power, the managers may pursue objectives other than the maximization of profit, such as the growth of the firm, of the staff, or the reduction of labor conflict by means of higher wages, at the expense of efficiency. Furthermore, the managers may devote resources to maintaining and increasing the levels of market power.

The positive relationship between market power and inefficiency is known as the "quiet life" effect. In the case of the banking sector, the only study that tests this hypothesis is Berger and Hannan (1998) for U.S. banks, which considers the relationship between cost efficiency levels and the market power of banking institutions. Nevertheless, this study presents the limitation of using market concentration (the Herfindahl index) as a proxy for market power. Thus, recent studies show the limitations of using market concentration measures as indicators of banking competition (Berger *et al.*, 2003; Maudos and Fernández de Guevara, 2004; Fernández de Guevara *et al.*, 2005a; Claessen and Laeven, 2004; among others).

In this context, the study has three fundamental aims: a) to quantify the level of market power in European banking sectors distinguishing different types of products (loans and deposits); b) to analyze the relationship between X-inefficiency in costs and the market power of the European banks; and c) to estimate the loss of welfare associated with market power. In the latter case, we analyze and quantify the two ways in which market power generates costs: the loss of net social welfare (Harberger's triangle) associated with the setting of prices above marginal costs, and the loss of efficiency in the management of the banks associated with the "quiet life" hypothesis.

Although we take as our starting point the study by Berger and Hannan (1998), the contributions of the study are as follows. Firstly, and in order to avoid the

limitations presented by the use of measures of concentration as indicators of competition, we use Lerner indices of market power. The advantage of the Lerner index over other indicators of competition (such as Panzar and Ross's test) is that it allows market power to be proxied at bank level and its evolution over time analyzed. Secondly, considering that banks can exercise different market power on either side of the balance sheet, we estimate indicators of market power separately for the markets of loans and deposits, this being the only study (as far as we know) that deals with this question in the case of European banks. Furthermore, unlike what is usual in the literature which deals with the measurement of market power, we estimate the marginal costs necessary for the calculation of the Lerner indices on the basis of a frontier costs function. And thirdly, we estimate the welfare losses associated with market power, both those related to the social loss from higher prices (welfare triangle), and the possible losses attributable to the cost inefficiency generated by the relaxation in bank management (quiet life effect). To estimate the social cost associated with market power, instead of making assumptions regarding the demand elasticities and market power as in Berger and Hannan (1998), we use a direct measure of the welfare loss following the methodology used in Oroz and Salas (2003), Fernández de Guevara et al. (2005) and Fernández de Guevara and Maudos (2004), where Harberger's welfare triangle is shown to be proportional to the Lerner index.

Results referring to the banking sectors of the EU15 show that although market power has decreased in the deposit market, the relative margins (Lerner index) have increased in the loan market. In the deposit market, margins are negative in the last years of the period analyzed, which suggests that banks follow a loss leader pricing strategy. We find a positive relationship between market power and cost X-efficiency, permitting us to reject the quiet life hypothesis. The social welfare loss attributable to market power in 2002 represented 0.54% of the EU15 GDP, with substantial variability across countries. Our results suggest that the welfare gains associated with a fall in market power may be far larger than the loss of bank cost efficiency driven by this lower level of market power. This fact shows the importance of the economic policy measures aimed at removing the barriers or obstacles that protect national markets from outside competition.

The rest of the paper is structured as follows. Section 2 reviews the literature on the relationship between market power and efficiency. Section 3 describes the approximation used in the estimation of market power, efficiency and the quantification of welfare loss from mis-pricing. Section 4 shows the results of the estimation of market power and the welfare triangle referring to the banking sectors of the EU15. Section 5 focuses on the relationship between efficiency and market power. Finally, section 6 contains the conclusions.

2. Market power and efficiency: background

This section reviews the theoretical and empirical background of the relationship between market power and efficiency. The literature on this issue is related to the hypotheses that explain the relationship between market structure and performance. In fact, the quiet life hypothesis is considered a special case of one of these hypotheses (the market power hypothesis).

In this context, there are three main hypotheses explaining the relationship between market structure and performance. The first one is the collusion hypothesis, also called the structure-conduct-performance (SCP) hypothesis (Bain, 1956). This hypothesis postulates that greater profits are the results of collusion between the firms of the industry. Thus, the SCP paradigm assumes that higher concentration enables banks to collude, which translates into extra profits.

The second one is the efficient structure hypothesis (Demsetz, 1973) which proposes an alternative explanation for the positive correlation between concentration and profitability, affirming that the most efficient banks obtain both greater profitability and market shares and, as a consequence, the market becomes more concentrated. In this case, the positive relationship observed between concentration and profitability is spurious and simply proxies for the relationship between superior efficiency, market share and concentration. More recently, Berger (1995) divided this hypothesis into the X-efficiency and scale efficiency hypotheses.

The third one is the relative market power hypothesis. Shepherd (1982 and 1986) establishes that the variance in performance is explained by efficiency as well as by the residual influence of the market share, because market share captures the influence of factors unrelated to efficiency, such as market power and/or product differentiation. Under this hypothesis, individual market share is the proxy variable for assessing market power.

The quiet life hypothesis can be considered a special case of the market power hypothesis. This hypothesis postulates that the higher market power, the lower the effort of managers to maximize operating efficiency, a negative correlation thus existing between market power and efficiency. Up to date, in the empirical testing of this hypothesis, market concentration measures are traditionally used as proxy for market power.

Berger and Hannan (1989) summarize the reasons that may explain the influence of market structure, as a proxy for market power, on efficiency. First, according to the quiet life hypothesis, if banks that compete in a market with higher concentration can set prices above marginal costs, managers do not have incentives to work as hard to keep costs under control. In other words, monopoly power allows managers to relax their efforts. Second, market power may allow managers to pursue objectives other than profit maximization (such as expense preference behavior). Third, in a non-competitive scenario, managers devote resources to obtaining and maintaining market power, which raises costs and reduces cost efficiency. And fourth, if banks enjoy market power, incompetent managers can survive without a wilful shirking of work efforts.

Only a few studies have analyzed the relationship between market power and efficiency in banking. For the U.S. banking industry, Berger (1995) implements tests that distinguish among the several hypotheses that explain the profit-structure relationship in banking using direct measures of efficiency. However, the relationship between efficiency and market power is not analyzed. Also for the U.S. banking sector, Berger and Hannan (1998) examine whether banks in more concentrated markets exhibit lower operating efficiency, and compare the efficiency cost of concentration with the loss measured by the welfare triangle. Results are consistent with the quiet life hypotheses and indicate that the efficiency costs estimated are much higher than the social cost occasioned by non-competitive pricing.

For the European banking sectors, although studies exist that test the hypotheses that may explain the profit-structure relationship (Molyneux *et al.*, 1994, 1996 and 2004; Goldberg and Rai, 1996; Maudos, 1998; Vander Vennet, 2002), as far as we know, there is no study that analyses the relationship between market power and efficiency. Therefore, to quantify this relationship, and to estimate the social loss derived from both market power and efficiency, are the main aims of this paper.

3. Methodology

3.1 The Lerner index of market power

The model most often used to obtain the Lerner index of market power in banking is the Monti-Klein imperfect banking competition model². This model examines the behavior of a monopolistic bank faced with a loan demand curve of negative slope $L(r_L)$ and a deposit supply of positive slope $D(r_D)$, the decision variables of the bank being L (volume of loans) and D (volume of deposits). For simplicity's sake the level of capital is assumed to be given and the bank is assumed to be price taker in the inter-bank market (r). As shown by Freixas and Rochet (1997), this model can be interpreted as a model of imperfect competition (Cournot) among a finite number of banks (N). Cournot's equilibrium is the set of N vectors (D_{n}^*, L_n^*) n=1,...,N which maximize the profit of bank n, considering the volume of deposits and loans of other banks to be given for each n. Thus, (D_{n}^*, L_n^*) is the solution of the following optimization problem:

$$\max_{(D_n,L_n)} \left\{ \left(r_L \left[L_n + \sum_{m \neq n} L_m^* \right] - r \right) L_n + \left(r - r_D \left[D_n + \sum_{m \neq n} D_m^* \right] \right) D_n - C \left(L, D \right) \right\}$$
(1)

where $C(\cdot)$ are the operating costs.

In equilibrium, each bank sets $D_n^* = D^*/N$ and $L_n^* = L^*/N$. From the first order conditions, we obtain:

$$\frac{\left[r_{L}^{*}-r-mc_{L}\right]}{r_{L}^{*}}=\frac{1}{N\varepsilon_{L}(r_{L}^{*})}\qquad\qquad\qquad\frac{\left[r-r_{D}^{*}-mc_{D}\right]}{r_{D}^{*}}=\frac{1}{N\varepsilon_{D}(r_{D}^{*})}\qquad(2)$$

where ε_L and ε_D are the elasticities of demand for loans and deposits respectively, and on the left hand side of each of equations (2) appears the expression of the Lerner index of market power for loans and for deposits, respectively. The Lerner index of market power defines the disparity between price (interest rate) and marginal cost (*mc*) expressed as a percent of price, taking into account that the divergence between product price and marginal cost of production is the essence of monopoly power. Thus, the Lerner Index measures the relative markup of price over marginal cost.

It is worth noting that according to expression (2), market power depends both on the elasticity of demand and on the number of firms competing in the market, which is usually proxied by measures of market concentration. Therefore, the advantage

² Monti (1972) and Klein (1971).

presented by the use of the Lerner index as a market power indicator is that it captures the influence both of market concentration and of demand elasticity, being therefore preferable to the use of market concentration indicators (such as CR(n), or the Herfindahl-Hirschman index).

The measurement of the Lerner index of market power requires prices and marginal costs to be estimated separately for loans and for deposits. In the first case, yearly averages of loan (deposit) interest rates for each bank were imputed from ratios of loan revenues (financial costs) to outstanding loan (deposit) values. In the second case, and bearing in mind that one of the objectives of the study is to analyze the relationship between X-efficiency and market power, marginal operating costs are estimated from a frontier translog cost function.

3.2 The welfare triangle

As Oroz and Salas (2003) and Fernández de Guevara *et al.* (2005) demonstrate, the Lerner index offers a proxy of the welfare loss due to market power. As figure 1 shows, assuming a linear loan (deposit) demand (supply) function and constant marginal cost, the loss in the consumer's surplus due to imperfect competition with respect to the situation of perfect competition is the area *acde* (*fihj*), while the gain in the producer's surplus is the area *abde* (*fghj*). Thus, the net social loss associated with misallocation of resources attributable to market power is the area *abc* (*ghi*), the socalled welfare or Harberger's triangle. This triangle shows the loss due to increasing the price from the competitive level to the monopoly price. If we express this social loss per unit of revenue (r_LL) – or unit of cost (r_DD) - the welfare triangle is proportional to the Lerner index³:

$$\frac{\Delta abc}{r_L^* L^*} = \frac{1}{2} \frac{r_L^* - r - mc_{L^*}}{r_L^*} \qquad \qquad \frac{\Delta gih}{r_D^* D^*} = \frac{1}{2} \frac{r - r_D^* - mc_{D^*}}{r_D^*}$$
(3)

The total welfare loss can be expressed as a percentage of GDP:

$$\frac{Welfare\ loss}{GDP} = \frac{1}{2GDP} \left[\frac{r_L^* - r - mc_{L^*}}{r_L^*} r_L^* L^* + \frac{r - r_D^* - mc_{D^*}}{r_D^*} r_D^* D^* \right]$$
(4)

³ Fernández de Guevera and Maudos (2004) adopt a different approach. They measure the welfare loss attributable to market power, adding the loss of consumer surplus.

3.3 X-Efficiency

Efficiency is measured using the concept of X-(in)efficiency and is regarded as a measure of the quality of management. The concept of cost efficiency measures the distance of a bank's cost relative to the cost of the best practice bank when both banks produce the same output under the same conditions.

We estimate X-inefficiency using the stochastic frontier approach proposed by Aigner *et al.* (1977) and Meeusen and van den Broeck (1977). This approach modifies the standard cost function by assuming that inefficiency forms part of the error term. Thus, the error term has two components. The first error component (v) is symmetric and captures the random variation of the frontier across firms, *i.e.* statistical noise, measurement error, and random shocks that are external to the firm's control. The second error component (u) is a one-sided variable that captures inefficiency relative to the frontier. Following Jondrow *et al.* (1982), bank-specific estimates of inefficiency terms can be calculated by using the distribution of the inefficiency term conditional to the estimate of the composite error term. We assume that the inefficiency term is drawn from a half-normal distribution. The resulting cost efficiently. For example, a bank with a cost efficiency of 0.85 is 85 per cent efficient or, equivalently, wastes 15 per cent of its costs relative to a best practice bank facing the same conditions (same output).

Taking into account that the aim of the paper is to analyze the quiet life hypothesis which relates market power to managers' efforts to control operating costs, the cost function we estimate excludes financial costs and, therefore, the price of deposits. Thus, the cost function, and hence the efficiencies estimated, include only operating expenses. If financial costs (and the price of deposits) were included in the cost function, efficiency cost would capture the effect of market power in the deposit market, and, since we aim to analyze the effect of market power on efficiency, it will bias our results.

We estimate a translog frontier cost function, where operating costs (*c*) depend on two outputs (*L*=loans, and *D*=deposits), two input prices (w_1 =price of labor and w_2 = price of physical capital) and technical change (Trend):

However, from a social point of view, the loss of consumer surplus is appropriated by banks in form of extraordinary profits.

$$\ln c_{it} = \sum \gamma_{h} \ln w_{hit} + \gamma_{L} \ln L_{it} + \gamma_{D} \ln D_{it} + \frac{1}{2} \sum \sum \gamma_{hm} \ln w_{hit} \ln w_{mit} + \gamma_{LD} \ln L_{it} \ln D_{it}$$

$$+ \frac{1}{2} \gamma_{LL} (\ln L_{it})^{2} + \frac{1}{2} \gamma_{DD} (\ln D_{it})^{2} + \sum \gamma_{hL} \ln w_{hit} \ln L_{it} + \sum \gamma_{hD} \ln w_{hit} \ln D_{it}$$

$$+ \mu_{I} Trend + \frac{1}{2} \mu_{2} Trend^{2} + \mu_{L} Trend \ln L_{it} + \mu_{D} Trend \ln D_{it}$$

$$+ \sum \mu_{h} Trend \ln w_{hit} + v_{it} + u_{it}$$
(5)

Observe that a time dummy variable is specified to capture the effect of technical change. Symmetry and linear homogeneity in input prices restrictions are imposed. According to this expression, operating marginal costs for loans and deposits are given by the following equations:

$$mc_{L_{it}} = \frac{\partial c_{it}}{\partial L_{it}} = \left[\gamma_L + \gamma_{LL} \ln L_{it} + \sum \gamma_{hL} \ln w_{hit} + \gamma_{LD} \ln D_{it} + \mu_L Trend\right] \frac{c_{it}}{L_{it}}$$
(6)
$$mc_{D_{it}} = \frac{\partial c_{it}}{\partial D_{it}} = \left[\gamma_D + \gamma_{DD} \ln D_{it} + \sum \gamma_{hD} \ln w_{hit} + \gamma_{LD} \ln L_{it} + \mu_D Trend\right] \frac{c_{it}}{D_{it}}$$

Cross-country comparison of cost efficiency requires estimation of a common cost efficiency frontier for all banks of the sample used. However, as some papers argue (DeYoung, 1998; Dietsch and Lozano-Vivas, 2000; Lozano *et al.*, 2002), when analyzing bank efficiency, it is important to allow for variation in environmental conditions which are beyond the control of bank managers. With this aim, we introduce into equation (5) some environmental variables (macroeconomic performance, economic development, population density, and banking services) that reflect how a bank's economic environment can help to explain cost efficiency differences among countries. More precisely, we control for the influence of:

-Per capita income, defined as the ratio of the Gross Domestic Product in real terms to the number of inhabitants (GDP/POP). This variable affects numerous factors related to the demand and supply of banking services (mainly deposits and loans). Additionally, this variable is used as an overall indicator of institutional development.

-The population density is measured by the ratio of inhabitants per square kilometer (POP/Km²). It is expected to have a negative influence on operating costs since high levels of population density should make retail distribution of banking services less costly, which should improve cost efficiency.

-Branches per capita (BR/POP) is an indicator of banking services. High levels of BR/POP imply high costs of providing banking services, which should reduce bank efficiency.

-Real GDP growth (GDPGR) is the annual rate of growth of GDP. This variable is introduced to capture the possible effect of the business cycle.

-Additionally, dummy variables for each country are included to take into account the influence of other remaining potential variables which are specific to each banking sector (e.g. regulatory and/or institutional variables).

4. Market power and welfare loss: empirical results

Bank data were obtained from BankScope (Bureau Van Dijk) database. The sample consists of a total of 29,744 observations of non-consolidated banking firms during the period 1993-2002. The banking sectors analyzed are those of the 15 countries of the European Union. The unbalanced panel data covers around 75% of bank assets included in the BankScope database in the European Union. It includes commercial banks, cooperative banks, savings banks and other types of institutions. Banks with missing data needed for estimating marginal costs and/or interest rates were not included. Additionally, banks with input prices and/or computed loan and deposit interest rates that were outside the interval of +/- 2.5 times the relevant standard deviation were dropped from the sample.

To estimate Lerner indices it is necessary to know the output prices that are included in the cost function. For this reason, and conditioned by the limited degree of dissagregation of the information contained in the data base used, the banking outputs considered were reduced to two broad aggregates from the balance sheet, one on the assets side and another on the liabilities. Thus, bank outputs are loans (proxied by total earning assets) and deposits (proxied by customer and short term funding). Input prices (*w*) are w1= price of labor (personnel costs / total assets⁴) and w2= price of capital (operating costs except personnel costs / fixed assets). The loan interest rate (r_L) is computed as the ratio of interest income and other operating income divided by loans. The deposit interest rate is proxied by the yearly average of the three month inter-bank deposit rate (reported by the Bank of Spain for the fifteen EU countries). The total

⁴ As data on the number of employees are not available in BankScope database, the price of labor is measured by the ratio of personnel expenses to total assets.

volume of loans and deposits for each country needed to compute the welfare triangle (see expression 4) is taken from the European Central Bank (2004).

Bank market concentration is measured by the Herfindahl-Hirshman index (HHI). For each country and year, concentration is computed using bank-level data from the BankScope database. Considering that concentration is a characteristic of the market, the HHI is computed as the squared sum of the market shares of all banks existing in a country in the whole BankScope database, not only in our sample. Environmental variables come from different statistical sources: (GDP/POP) and GDPGR are obtained from the OECD's National Accounts; (BR/POP) comes from Bank Profitability (OECD) completed with the European Central Bank's Blue Book on payment statistics and National Accounts (OECD); (POP/Km²) is obtained from National Accounts (OECD) and NewCronoss (Eurostat). Table 1 contains the descriptive statistics of the variable used in the paper.

The evolution of interest rates, marginal costs and absolute margins are reported in table 2. Loan and deposit interest rates have decreased in all countries, in a context of reduction of inflation and nominal convergence among countries. Nominal convergence has accelerated in recent years as a consequence of the adoption of the euro in the European Monetary Union. Estimated marginal costs of loans are higher than those of deposits with relatively stable behavior in the period analyzed. Deposit marginal costs have decreased in almost all countries (with the exception of Austria, Belgium and Greece).

The evolution of loan $(r_L - r - mc_L)$ and deposit $(r - r_D - mc_D)$ absolute margins are shown in the last column of table 2. Loan margins rose over the period in the majority of countries (from 0.7% in 1993 to 2.1% in 2002 for the average of the EU15) while deposit margins fell. This suggests that market power may have increased in loan markets while falling in deposit markets. In recent years (1997-2002), margins were negative in the deposit market, suggesting a loss leader pricing strategy: although deposits may not be profitable by themselves, they allow banks to capture customers, and banks can exercise market power in the loans market.

Table 3 shows the evolution of the Lerner index of market power for each of the EU15 countries, and for the EU15 weighted average. Market power has increased in the loans market in all the European banking sectors with the exception of Sweden and the United Kingdom. In 2002, the highest values of the index correspond to Luxembourg (0.51) and Portugal (0.46), which together with France, Germany, Greece, Italy, Netherlands and Spain are situated above the average of the EU15 (0.32). At the

opposite extreme are the UK and Sweden, with low levels of market power in the loans market.

In deposits, market power has decreased in all countries except in Finland, Sweden and the UK. Furthermore, from the mid-1990s, the Lerner index is negative in almost every country. The fact of finding negative margins in the deposit market is not driven by the level of marginal operating costs, but by the spread between the money market rate and the deposit interest rate $(r - r_D)$, which is negative in most European banking sectors. As we have mentioned above, this suggests a loss leader pricing strategy in the deposit market⁵. Furthermore, the negative margins in deposits also reflect the competition of other financial liabilities such as mutual funds. On the assets side, loans remain the principal source of finance for certain segments, especially where relationships are more important (small firms or households), being, therefore, less subject to the competition of other financial intermediaries or markets.

With this estimation of the Lerner index as a base, we can compute the social loss due to misallocation of resources attributable to market power from expression (4). Our measure comprises the loss of consumer and producer surplus occasioned by noncompetitive pricing (the welfare triangle). Table 4 shows the welfare loss associated with imperfect competition as a percentage of GDP. The evolution of the EU15 average suggests that in the period between 1993 and 2002, there was a downward trend in the welfare loss attributabled to market power broken down in the last years. Whereas in 1993 the social loss of market power was 0.54% of GDP, it decreased to a 0.27% of GDP in 2000, showing a sharp increase since then, and recovering the initial level of 0.54% of GDP in 2002. Given the differences in the evolution of market power of loans and deposits, the evolution of the welfare loss is due to the combination of market power exercised in the loan and deposits markets, counteracting the effects of the latter on the former until 2000. So, it seems that the loss leader strategy adopted by European banks by which banks set interest spreads under their marginal costs in the deposits markets -so that they can establish a relationship with the client which allows banks to exercise market power in loans-, was not only profitable for banks, but also for society as a whole given the reduction of social losses until 2000. Since then the effects of greater market power in the loan markets have dominated generating higher social losses.

⁵ For the Spanish case, results are similar to those reported in Carbó *et al.* (2005), who analyse the intensity of price and non-price (branches) competition over the 1986-2002 period. Their results show that market power has increased in the loan markets while

The data also show that there is substantial variability in the indicator across countries. Thus, in 2002, and apart from Luxembourg - whose high social efficiency is driven by the importance of the banking sector in the economy⁶ - countries with welfare losses over 1% of GDP such as Denmark (1.1), Spain (1.4), Portugal (1.6) and Ireland (1.8) coexist with countries with losses below 0.5% such as United Kingdom (0.4), Netherlands (0.3), Belgium (0.3) and specially France (0.03). When interpreting these results, one must note that the value and the evolution of the social inefficiency of banking intermediation depend on two factors (see expression 4): the evolution of market power (Lerner index) and the evolution of the ratio of banking assets (loans and deposits)/GDP. Thus, in some countries the magnitude of the welfare loss is due more to the banking orientation of the financial structures in each country rather than to the market power of banks. Furthermore, it is of concern to observe a high degree of bank market power in countries that are strongly orientated towards bank financing, such as Ireland, Portugal and Spain.

The estimates of the welfare loss triangle are lower than the values reported by Fernández de Guevara and Maudos (2004) for the period 1993-2000. Thus, taking as reference their estimation for 2000, the loss of welfare associated with imperfect competition is 2.51% of GDP for the average of the EU15, as against the 0.27% obtained in this study. Several factors can help to explain these differences. Firstly, the welfare loss estimated in Fernández de Guevara and Maudos (2004) incorporates, as well as Harberger's welfare triangle, the loss of consumer surplus that is appropriated by the producer in form of extra profits. Secondly, Fernández de Guevara and Maudos (2004) use a single indicator of banking activity (proxied by total assets) and, consequently, a single indicator of market power. Taking into account that the levels of market power are very different in the loan and deposit markets, the use of a single indicator of market power could bias the estimated value of welfare loss.

falling in deposit markets. They also found that in the 1997-2002 period, margins were negative in the Spanish deposits market.

⁶ According to the European Central Bank (ECB, 2004), in Luxembourg, the ratio of deposits (from non-credit institutions) / GDP in 2002 was 9 whereas in loans (to non-credit institutions) "only" 5, thus dominating the effect of the negative Lerner index in the deposit market power over the degree of market power in the loan market.

5. The relationship between market power and efficiency: empirical results

In this section we present the results regarding the relationship between market power and efficiency in the European banking sectors. Having analyzed the evolution of market power in section 4, we now turn to the study of cost efficiency.

Table 5 shows the mean efficiency scores, estimated by the stochastic frontier approach. Average cost efficiency is quite stable and ranges around an average level of 85%, which is in accordance with former studies of the EU (Maudos *et al.*, 2002; Altunbas *et al.*, 2001)⁷. If we take as reference the last year analyzed, Finland (93.2%), Ireland (91.9%) and the United Kingdom (89.9%) are the most efficient banking sectors, while the most inefficient are Portugal (74%), Belgium (77.6%) and Luxembourg (80%). If we focus on the evolution over time, there are important differences among countries, efficiency cost increasing in seven of the EU15 countries.

Just as the existence of market power implies a loss of welfare, the inefficiency of banking institutions is also a cost for society. According to Maudos and Fernández de Guevara (2004), banks establish their margins as a function of the operating costs they have to bear. Thus, these authors obtain a high elasticity of banking margins against levels of efficiency. It can thus be understood that in the final instance, it is the consumer of banking services that have to pay the costs of the operating inefficiency of banking institutions. We can, therefore, compare the social loss of market power with the costs for society represented by the inefficiency of the banks. Table 6 shows, as a percentage of GDP, the difference between the minimum costs of production that define the frontier of efficient behavior, and the effective costs incurred by European banking institutions. For the European Union as a whole, the social cost of bank inefficiency increased slightly from 0.34% in 1993 to 0.4% of GDP in 2002 as was to be expected in view of the slight downturn of cost efficiency levels during the period analyzed. By countries, it can be seen that the highest losses associated with inefficiency are found in Luxembourg (2.4%), Belgium (1.1%), Portugal (0.6%) and France (0.5%), while in 2002 the lowest losses of efficiency are observed in Ireland (0.007%), Finland (0.1%), United Kingdom (0,12%) and Sweden (0.16%).

If we compare the levels of social welfare loss derived from market power (table 4) with the magnitude of welfare loss associated to banks' cost inefficiency (table 6), we can see that, in general, the social welfare losses derived from market power are greater

⁷ Averaging the results of 130 studies across five different types of frontier approaches for 21 countries suggests that average cost inefficiency in various nations' banking industries is 20% to 25%. See Berger and Humphrey (1997).

than those from X-inefficiency. Thus, in 2002, the former was 0.54% of GDP, in contrast to the 0.4% of X-inefficiency⁸. On average, over the period, social losses derived from market power are a 23% higher than that of X-inefficiency.

Once we have computed the social loss from both the level of market power and cost efficiency we focus on the relationship between them. According to the quiet life hypothesis, part of the level of inefficiency is caused by market power. To test the validity of the quiet life hypothesis for the European banks, we estimate an equation where the dependent variable is cost efficiency and the independent variables are the Lerner index and other variables that potentially influence bank efficiency. These factors include both bank and country characteristics that may be associated with managerial decisions. Specifically, the explanatory variables of cost efficiency are the following:

-Concentration: traditionally, studies use market concentration measures as proxy variables for market power. For that reason, we also use market concentration (HHI) as a first proxy variable for competition. Since no information is available on banking activities in European local markets, our measures of concentration are calculated at a national level. Initially, the HHI is measured in terms of total assets (HHIA) to check the similarities with the results of Berger and Hannan (1998). Alternatively, taking into account the evidence offered before, and the results of Corvoisier and Gropp (2002) and Fernández de Guevara et al. (2005) in which the effect of concentration is different in different banking products, the HHI is used in terms of loans (HHIL) and deposits (HHID). However, some recent papers show the limitations of using concentration measures to proxy for the competition environment in banking markets (Berger et al., 2003; Fernández de Guevera et al. 2005a; Claessens and Laeven; 2004). Moreover, the relationship between the level of competition and concentration is not straightforward and depends on the conduct of the banks in the market. As Shaffer (2004) shows, in a concentrated market different competitive equilibria can be obtained depending on the conduct of banks in markets. Consequently, as our preferred alternative to concentration ratios, we use the Lerner index as a direct measure of competition.

-Market power: to test the effect of market power on managers' efforts to control operating costs, the Lerner index is introduced as a determinant of cost efficiency. A negative influence of cost efficiency would favor the quiet life hypothesis.

⁸ An exceptional case is that of France, where the high level of competition causes losses of cost efficiency being greater than those of market power.

-Size: the size of each bank, measured by total assets (TA), is included to test if larger banks are able to get better management than smaller ones. Berger and Humphrey's (1997) review article concludes that there is consistent evidence that larger banks tend to be more efficient than small ones.

-Specialization (S): we distinguish four types of banking specialization: commercial bank (S1), savings bank (S2), cooperative bank (S3), and others (investment bank, medium-and long term credit bank, real estate/mortgage bank, specialized government credit institution, etc). Dummy variables (which take the value 1 when the bank adopts specialization and 0 otherwise) are used to control for the possible influence of institutional specialization on cost efficiency⁹.

The regression model is the following:

$$EF_{it} = f(Market \ power_{it}, \ln(TA_{it}), S)$$
(7)

where i=bank, t=year, EF is the level of efficiency and market power is, alternatively, the HHI concentration index or the Lerner index.

Since operating cost efficiency (our dependent variable) is a variable bounded between zero and one, it is necessary to use a non-linear specification of the functional form f, rather than a linear regression model. Using in equation (7) the logistic functional form

$$logistic(EF_{it}) = \frac{e^{EF_{it}}}{1 + e^{EF_{it}}} = \frac{e^{\beta_1 Market \ power_{it} + \beta_2 \ln(TA_{it}) + \sum_{c=1}^4 \beta_3 S_c}}{1 + e^{\beta_1 Market \ power_{it} + \beta_2 \ln(TA_{it}) + \sum_{c=1}^4 \beta_3 S_c}}$$
(8)

which can be easily linearized via the logit transformation as follows,

$$\ln\left[\frac{\text{logistic}(EF_{it})}{1 - \text{logistic}(EF_{it})}\right] = \beta_1 Market \ power_{it} + \beta_2 \ln(TA_{it}) + \sum_{c=1}^4 \beta_{3c} S_c$$
(9)

Table 7 reports the results of the estimation of expression (9). This equation was estimated introducing both individual fixed effects and time effect. Column (1) shows results using market concentration in terms of total assets (HHIA) as a proxy variable for market power. The coefficient of the HHA is negative but not statistically different from zero. The sign -although not the significance- is similar to the result of Berger and Hannan (1998) showing evidence in favor of the quiet life hypothesis. However, if we introduce the effect of concentration in the loan and deposit market (column 2), the

⁹ S4 (other type of institutions) is the group of reference in the estimations.

influence is positive and statistically significant in the loan market and negative in the deposit market. This result implies that operating in more concentrated markets is associated with less cost efficiency in the deposit market but with more efficiency in the loan market. As in Corvoisier and Gropp (2002) and Fernández de Guevara *et al.* (2005), this result shows the importance of distinguishing the effect of concentration by type of product, as done in this paper. Thus, the results, though they support the quiet life hypothesis in the deposit market, allow us to reject this hypothesis in the loan market, as banks in more concentrated loan markets exhibit higher (and not lower) cost efficiency. This result must be taken with caution. As we have mentioned before, concentration indices have been calculated at national level, whereas by the nature of banking services, competition takes place at a lower than national level.

Table 7 also shows the elasticities of cost efficiency w.r.t. changes in the explanatory variables. In terms of economic impact, a 100% increase in the HHI in the loan (deposit) market would cause efficiency to increase (decrease) by 0.99% (0.86%). Thus, an increase in loan market concentration from that of the least concentrated European loan market (HHL=180.42 in Germany in 2002) to that of the most concentrated market (HHL=2,384.4 in Finland), would increase cost efficiency by 9%. In the deposit market, an increase in market concentration from the minimum value corresponding to Germany (188.28) to the maximum of the Netherlands (3,196.2), would decrease efficiency by 12%.

Regarding the rest of the explanatory variables of cost efficiency, size has a positive and statistically significant influence on efficiency, indicating that larger banks are more cost efficient. Its economic impact is larger than that of concentration, showing an elasticity of 1.85. Thus, if a bank duplicates its size (a 100% increase), its cost efficiency would increase by 1.85%. With increasing returns to scale, greater size may increase bank efficiency through more efficient scale. The type of banking specialization does not seem to influence banks' efficiency, since none of the dummy variables representing specialization is statistically significant.

As we have mentioned above, recent studies have shown the limitation of using market concentration measures to proxy for the degree of competition in the markets. For that reason, and because it is impossible for us to calculate HHI indices at a lower than national level, column (3) of table 7 shows results using the Lerner index as proxy for competition. In this case, the results show that the coefficient of the Lerner index, in the loan market as well as in the deposit market, is statistically positive (at 1% level), which indicates the existence of a negative relationship between competition and cost efficiency in the European banking sectors. In terms of economic impact, a 100% increase in the Lerner index in the loan (deposit) market would produce an increase of

0.29% (0.05%) in cost efficiency. In respect of the influence of the rest of the explanatory variables of efficiency, the results are similar to those corresponding to column (2).

The difference of results obtained, in terms of the impact on efficiency, between market concentration and Lerner index, shows the existence of a low relationship between the two variables. In fact, the correlation coefficient between the Lerner index and the HH index is -0.07 in the loan market and 0.07 in the deposit market, and is not statistically significant in either case. Therefore, the absence of correlation between the Lerner index and the HH index, as well as their different influence on cost efficiency, shows the limitations of market concentration measures as proxy variables for competition.

These results permit us to reject the quiet life hypothesis in the European banking system. There may be several reasons explaining the positive effect of market power on efficiency. Firstly, as Pertersen and Rajan (1995) argue, banks with monopolistic power due to their location (close to the firms) have lower costs of monitoring and transacting with firms¹⁰. Secondly, banks that possess market power due to geographical or technological specialization may have cost advantages in screening certain groups of borrowers (Kaas, 2004). Thirdly, market power allows banks to enjoy greater profits, which may create incentives to behave prudently (enhancing bank stability). This more prudent behavior leads to the selection of less risky activities with lower costs of monitoring, therefore increasing cost efficiency. And fourthly, the banks that enjoy greater market power are under less pressure to increase the quality of banking services (less availability of means of payment, worse attention to customers, etc.), thus decreasing operating cost and increasing their cost efficiency.

Having reached this point, it is of interest to value the effects of economic policy measures aimed at increasing competitive rivalry in European banking markets. This poses the following questions. Firstly, what would be the gain in terms of social welfare if market power decreased in the European banking sectors where there is least competitive rivalry? But in the light of the results obtained, this hypothetical reduction of market power would generate an increase in X-inefficiency that increases the banking margins borne by the consumers of banking services (Maudos and Fernández de Guevara, 2004). Secondly, therefore, what would be the potential loss of welfare

¹⁰ The theoretical contribution of Sussman and Zeira (1995) show that banks' monitoring costs increase in distance.

associated with the cost inefficiency of European banks, given the positive relationship found between market power and cost efficiency?

The reduction of cost efficiency associated with a decrease of market power can be quantified using the regression coefficient of the Lerner index (table 7, column 3) and the evolution of the values of the Lerner index (table 2). Once this loss has been quantified, the second step is to compare its magnitude with the reduction of the social welfare loss associated with less market power. If the increase in social welfare is greater than the loss of cost efficiency, economic policy initiatives must be aimed at reducing the market power of the banks, as the cost of such policies in terms of loss of cost efficiency (which would be translated into higher banking margins) is lower than the gains in the social efficiency of financial intermediation.

To be able to carry out this exercise it is necessary to make some assumptions as to the evolution of market power and thus to quantify its impact on welfare and the level of efficiency. Specifically, we will assume that market power in the credit and deposit markets decreases up to the observed EU15 average in those banking sectors in which the value of the Lerner index is greater than this average, remaining at the current values in the other countries.

Table 8 shows the reduction of the welfare loss triangle associated with the assumption of reduction in market power. In the table, the social welfare gain is quantified as a percentage of the GDP for each country and year of the sample, as well as for the weighted average of the EU15. If we focus on the last year analyzed (2002), the social welfare gain due to the decrease in market power represents 0.32% of GDP. With the exception of Luxembourg (whose high gain is explained by the high relative importance of the banking sector), the gains vary between values higher than 0.5% of GDP (0.6 in Spain, 0.8% in Portugal, and 0.9 in Ireland and United Kingdom) and values lower to 0.2% (Belgium, Denmark, Italy, Germany and France).

The cost efficiency loss associated with the reduction of market power in the banking sectors with Lerner indices above the EU average appears in table 9. The elasticities of efficiency against changes in the Lerner index shown in table 7 are used to calculate the impact on efficiency of the variation in the level of competition in loans and deposits markets. As a percentage of GDP, the increase in cost inefficiency is only 0.0014% for the average of the EU15 in 2002, the loss being very small in practically all European countries. The only exception is the loss of efficiency of Luxembourg (0.018%) as a consequence of the importance of banking assets.

The comparison of the results of tables 8 and 9 shows that the welfare gains associated with the fall of market power are much greater than the loss of bank cost efficiency. Consequently, from a social point of view, the economic policy measures adopted by the different European institutions with banking competencies (European Commission, European Parliament, European Central Bank, national Ministries of Economy, Antitrust authorities, etc.) must be aimed at eliminating all kinds of barriers or obstacles to banking competition, given the magnitude of the potential effect of such measures on social welfare. In this respect, we might highlight the current Financial Services Action Plan of the European Commission, which incorporates 42 measures aiming to create a single European financial market.

For the U.S. banking industry, the results of Berger and Hannan (1998) support the quiet life hypothesis: the additional operating costs attributable to market power appear to be several times larger than the social loss due to the non-competitive pricing of bank outputs, as measured by the welfare triangle. As well as the difference in the sample used (U.S. *vs.* European banks), there are two main reasons that may justify the discrepancy with the evidence that we have obtained in this study. Firstly, we use the Lerner index as market power indicator, given the advantages that it presents over the use of market concentration ratios, especially when market concentration indices cannot be calculated at local level. And secondly, our empirical approach to the measurement of the welfare triangle loss does not need to make assumptions as to the proportional increase in price owing to the exercise of market power or as to demand price elasticities.

6. Conclusions

This paper estimates the two paths by which market power affects social welfare. On the one hand, greater market power implies a loss of social welfare (the socalled welfare triangle). On the other hand, market power may influence the efforts of managers to control costs and, consequently, cost efficiency (quiet life hypothesis). With this aim, the paper examines the relationship between market power (proxied by the Lerner index) and cost efficiency (proxied by X-efficiency measures) in banking. Furthermore, this paper is the first one to estimate Lerner indices of market power for different banking products, namely loans and deposits, in the European Union banking sectors.

The results referring to the banking sectors of the European Union-15 over the period 1993-2002, show that while market power increased in the loan market, it

decreased in the deposit market. The results also show that margins are negative in the deposit market, suggesting that banks follow a loss leader pricing strategy. The welfare loss (Harberger's triangle) from the misallocation of resources attributable to market power represented 0.54% of the GDP of the European Union in 2002, substantial variability existing across countries.

Given the limitations presented by the use of market concentration measures as proxy variables for bank competition, we tested the quiet life hypothesis by examining the relationship between cost efficiency and the Lerner index. The results show the existence of a positive relationship between market power and cost efficiency, permitting the rejection of the quiet life hypothesis. The lower pressure from competition to increase the quality of banking services, together with lower monitoring and screening costs, may explain the negative relationship between competition and efficiency.

Although a reduction in market power decreases the size of the social loss from mispricing of bank outputs, it decreases the cost efficiency of the banking system, posing the question of its net impact for society. The simulation carried out under the assumption that the degree of competition increases in the European banking sectors that enjoy a level of market power above the average for the European Union shows that the welfare gains associated with the fall of market power are much greater than the loss of bank cost efficiency. This result shows the relevance of economic policy measures (such as the Financial Services Action Plan of the European Commission) aimed at removing the barriers or obstacles that protect national markets from outside competition (different taxation of banking products among countries, differences in regulation and supervision, entry barriers that hinder cross-border banking penetration, etc).

Finally, the discrepancy of our results from those obtained by Berger and Hannan (1998) for the U.S. banking industry show the need for additional empirical evidence referring to other banking sectors, because, as far as we know, there are hardly any studies that estimate the costs of market power, whether in terms of social welfare or in terms of bank cost efficiency.

References

Aigner, A., Novell, C.A.K. and Schmidt, S. (1977). "Formulation and estimation of stochastic production function models", *Journal of Econometrics* 86, 21-37.

Altunbas, Y., Gardener, E. P. M., Molyneux, P. and Moore, B. (2001). "Efficiency in European banking", *European Economic Review* 45(10), 1931-1955

Bain J. (1956). *Barriers to new competition*, Cambridge. Mass. Harvard University press.

Berger, A. (1995). "The profit-structure relationship in banking –test of market power and efficienct-structure hypotheses", *Journal of Money, Credit and Banking* 27(2), 404-431.

Berger, A., and Humphrey, D. (1997). "Efficiency of Financial Institutions: International Survey and Directions for Future Research", *European Journal of Operational Research* 98, 175-212.

Berger, A.N. and Hannan, T.H. (1998). "The efficiency cost of market power in the banking industry: a test of the 'quiet life' and related hypotheses", *Review of Economics and Statistics* 8(3), 454-465.

Berger, A., Demirgüç-Kunt, A., Levine, R. and Haubrich, J. (2003). "Bank concentration and competition: an evolution in the making", *Journal of Money, Credit and Banking* 35(5), 433-451.

Bikker, J.A. and Haaf, K. (2002). "Competition, concentration and their relationship: an empirical analysis of the banking industry", *Journal of Banking and Finance* 26, 2191-2214.

Carbó, S., Fernández de Guevara, J., Humphrey, D. and Maudos, J. (2005). "The intensity of price and non-price competition in banking", Documento de Trabajo DT 2-05, Fundación BBVA, Bilbao.

Claessens, S. and Laeven, L. (2004). "What drives bank competition? Some international evidence", *Journal of Money, Credit and Banking* 36(3), 563-583.

De Bandt, O. and Davis, E. P. (2000). "Competition, contestability and market structure in European banking sectors on the eve of EMU", *Journal of Banking and Finance* 24, 1045-1066.

Demsetz, H. (1973). "Industry Structure, Market Rivalry, and Public Policy", *Journal of Law and Economics* 16, 1-9.

DeYoung, R. (1998). "X-Inefficiency and Management Quality in Commercial Banks", *Journal of Financial Services Research* 13, 5–22.

Dietsch, M. and Lozano-Vivas, A. (2000). "How the Environment Determines the Efficiency of Banks: A Comparison between French and Spanish Banking Industry" *Journal of Banking and Finance* 24, 985–1004.

ECB (2004): *Report on EU banking structure*, European Central Bank, Frankfurt, November.

Fernández de Guevara, J. and Maudos, J. (2004). "Measuring welfare loss of market power: an application to European banks"; *Applied Economics Letters* 11(13), 833-836.

Fernández de Guevara, J., Maudos, J. and Pérez, F. (2005). "Market power in European banking", *Journal of Financial Services Research* 27(2), 109-138.

Fernández de Guevara, J., Maudos, J. and Pérez, F. (2006) "Integration and competition in European financial markets", *Journal of International Money and Finance*, forthcoming.

Freixas, X. and Rochet, J-Ch. (1997). Microfoundations of Banking, The MIT Press.

Goldberg, L. G. and Rai, A. (1996). "The Structure-Performance Relationship for European Banking", *Journal of Banking and Finance* 20, 745-771.

Jondrow, J., Novell, C.A.K., Materov, I.S and Schmidt, P. (1982). "On the estimation of technical inefficiency in the stochastics frontier production models", *Journal of Econometrics* 19, 233-38.

Kaas, L. (2004). "Financial market integration and loan competition. When entry deregulation is socially beneficial?" *European Central Bank*, Working Paper series No. 403, November.

Klein, M. (1971). "A theory of the banking firm", *Journal of Money, Credit and Banking* 3, 205-218.

Lozano-Vivas, A., Pastor, J. and Pastor, J.M. (2002). "An Efficiency Comparison of European Banking Systems Operating under Different Environmental Conditions", *Journal of Productivity Analysis* 18, 59–77.

Maudos, J., Pastor, J.M., Pérez, F. and Quesada, J. (2002). "Cost and profit efficiency in European banks", *Journal of International Financial Markets, Institutions and Money* 12, 33-58.

Maudos, J. and Fernández de Guevara, J. (2004). "Factors explaining the interest margin in the banking sectors of the European Union", *Journal of Banking and Finance* 28(9), 2259-2281.

Meeusen; W. and van den Broeck, J. (1977). "Efficiency estimation from Cobb-Douglas production function witj componed error", *Internacional Economic Review* 18, 435-44.

Molyneux, P., Lloyd-Williams and Thornton, J. (1994). "Competitive conditions in European banking", *Journal of Banking and Finance* 18, 445-459.

Molyneux P., Y. Altunbas and Gardener, E. (1996). "The Structure-Performance Relationship in US and European Banking" in *Efficiency in European Banking*, John Wiley & Sons Ltd. pp. 93-136.

Molyneux, P., Goddard, J. and Wilson, J. (2004). "Dynamics of growth and profitability in banking", *Journal of Money, Credit and Banking* 36(6), 1069-1090.

Monti, M. (1972). "Deposit, credit, and interest rate determination under alternative bank objetives", in G.P. Szego and K. Shell (eds.), *Mathematical methods in investment and finance*, Amsterdam, North-Holland.

Oroz, M. and Salas, V. (2003). "Competencia y eficiencia en la intermediación financiera en España: 1977-2001", *Moneda y Crédito* 217, 73-99

Petersen, M.A. and Rajan, R.G. (1995). "The effect of credit market competition on lending relationship", *Quartely Journal of Economics* 110, 407-443.

Shaffer, S. (2004). "Patterns of competition in banking", *Journal of Economics and Business*, 56(4), 287-313.

Shepherd, W. (1982). "Economies of scale and monopoly profits", in *Industrial Organization, Antitrust, and Public Policy*, J.V. Craven (ed.), Boston, Kulwer Nihoff.

Shepherd, W. (1986). "Tobin's q and the structure-performance relationship: reply", *American Economic Review* 76, 1205-10.

Sussman, O., and J. Zeira (1995). "Banking and Development," CEPR Discussion Paper 1127.

Vander Vennet, R. (2002). "Cost and Profit Efficiency of Financial Conglomerates and Universal Banks in Europe", *Journal of Money, Credit and Banking* 34(1), 254-282.

Table 1. Sample statistics of variables

	с	L	D	w1	w2	r_L	r_D	r	GDPGR	GDP/POP	BR/POP	POP/Km ²	HHIA	HHIL	HHID	LA	BANKS
									1993								
AUSTRIA	1.54	95.12	77.88	0.95	88.56	9.18	6.81	6.77	0.42	19.99	0.59	0.10	687.00	741.81	812.51	6.68	57
BELGIUM	1.46	95.77	85.98	0.92	74.99	9.57	7.34	8.09	-0.96	18.72	1.97	0.30	838.93	1,057.49	884.82	6.85	62
DENMARK	2.35	95.14	84.71	1.44	50.72	12.46	6.78	10.08	0.00	24.65	0.45	0.12	1,442.12	1,741.50	2,232.22	6.22	70
FINLAND	2.08	88.73	73.01	0.86	72.44	9.92	7.11	7.55	-1.24	17.48	0.43	0.01	1,854.48	2,489.68	2,588.74	7.08	4
FRANCE	1.68	92.86	80.71	0.95	102.49	9.66	7.17	8.28	-0.89	18.73	0.45	0.11	399.45	431.35	507.48	6.97	264
GERMANY	1.62	96.59	75.82	1.00	73.22	9.50	6.79	7.19	-1.09	20.24	0.55	0.23	197.41	213.89	172.63	6.38	1,164
GREECE	2.12	91.43	85.37	1.50	47.60	14.63	12.23	22.90	-1.60	8.71	0.11	0.08	2,142.90	3,137.13	1,797.41	6.63	17
IRELAND	2.25	95.90	88.33	1.16	40.52	12.90	6.44	8.61	2.69	14.36	-	0.05	1,787.41	2,099.22	1,862.45	6.34	5
ITALY	2.77	90.55	79.62	1.66	50.24	13.13	7.69	10.11	-0.88	16.29	0.37	0.19	455.57	309.57	356.56	6.67	222
LUXEMBOURG	0.86	96.75	88.97	0.45	72.98	9.58	8.37	8.09	4.20	33.26	0.77	0.15	291.79	359.22	303.42	6.36	72
NETHERLANDS	2.03	95.10	86.65	0.83	97.80	8.76	5.98	6.73	0.65	19.04	0.47	0.37	1,651.48	1,962.55	2,413.53	6.20	19
PORTUGAL	2.48	84.13	85.12	1.40	42.58	16.40	9.08	12.63	-2.04	8.39	0.29	0.11	916.83	700.30	846.87	6.54	28
SPAIN	2.55	92.74	88.56	1.56	35.20	14.17	7.95	11.69	-1.03	11.31	0.90	0.08	517.71	463.24	459.24	6.75	115
SWEDEN	2.25	92.85	77.29	1.02	231.61	13.26	8.41	8.50	-2.00	21.61	0.32	0.02	1,198.07	1,210.24	1,938.01	7.28	6
UK	2.27	91.99	83.01	1.26	88.25	9.79	5.69	5.71	2.33	19.92	0.20	0.24	663.83	754.90	738.82	6.99	90
									2002								
AUSTRIA	1.73	95.39	77.68	0.94	68.23	6.52	3.36	3.27	1.37	24.15	0.53	0.10	415.42	367.80	378.59	6.19	133
BELGIUM	1.50	91.48	82.18	0.76	77.81	6.10	3.56	3.30	0.70	22.40	1.08	0.31	1,601.10	1,645.31	1,678.35	7.38	34
DENMARK	1.41	86.97	67.59	0.86	90.85	6.15	3.17	3.30	1.02	30.19	0.38	0.12	1,228.52	1,111.92	2,176.85	6.64	70
FINLAND	1.55	92.89	79.86	0.75	129.09	6.39	2.61	3.21	2.27	23.83	0.25	0.02	2,604.78	2,384.40	2,867.61	7.53	4
FRANCE	1.44	90.09	72.39	0.81	107.54	6.64	5.35	3.30	1.18	22.07	0.38	0.11	498.84	489.94	486.32	7.29	199
GERMANY	1.29	96.46	76.57	0.69	90.56	6.48	4.40	3.28	0.08	22.88	0.43	0.23	199.88	180.42	188.28	6.52	1,329
GREECE	2.27	95.37	89.94	1.38	55.90	7.74	2.92	3.30	3.63	11.20	0.28	0.08	1,715.03	1,445.92	1,762.55	7.00	14
IRELAND	0.70	98.43	91.89	0.32	46.42	6.15	3.12	3.30	6.13	27.21	0.28	0.06	742.49	872.14	821.98	6.57	4
ITALY	2.21	90.00	64.39	1.22	87.19	7.82	3.51	3.30	0.36	18.93	0.51	0.19	352.20	316.73	302.56	6.57	486
LUXEMBOURG	0.80	95.16	78.18	0.41	83.63	8.03	7.96	3.30	2.47	45.72	0.62	0.17	403.07	717.46	427.56	6.78	57
NETHERLANDS	1.13	96.46	66.40	0.64	73.40	6.87	5.74	3.30	0.57	23.41	-	0.39	1,500.56	1,207.29	3,196.21	6.65	15
PORTUGAL	1.70	88.20	79.43	0.74	80.54	8.37	4.67	3.29	0.38	10.47	0.48	0.11	1,377.50	1,572.55	1,607.46	6.88	22
SPAIN	1.78	91.88	79.20	1.07	47.77	7.48	2.85	3.31	2.04	14.54	0.95	0.08	599.12	513.99	565.94	7.03	112
SWEDEN	1.58	87.87	76.31	0.83	328.52	6.91	3.53	4.12	2.10	27.66	0.21	0.02	1,078.44	1,046.96	1,600.39	6.42	91
UK	1.50	93.31	78.69	0.81	145.32	5.12	2.65	3.97	1.64	25.22	0.18	0.24	408.45	555.27	482.30	7.09	106

Banks of Ireland in 1993 and 1994 and the Netherlands in 2001 and 2002 have not been included in the sample for the estimation of the cost function because the number of branches was not sample size in the cost function is 29,694 observations.

sample size in the cost function is 29,694 observations. c = Operating costs (% of total assets). Source: BankScope (Bureau Van Dijk) L=Loans (total earning assets) (% of total assets). Source: BankScope (Bureau Van Dijk) D=Deposits (customer and short term funding) (% of total assets). Source: BankScope (Bureau Van Dijk) w1=Price of labor (Personnel expenses / Total Assets) %. Source: BankScope (Bureau Van Dijk)

w2 =Price of capital (Non-personnel operating expenses / Fixed assets) %. Source: BankScope (Bureau Van Dijk) r_L = Loans' interest rate (Interest income and other operating income / L) %. Source: BankScope (Bureau Van Dijk)

r_D=Deposits' interest rate (Interest expenses / D) %. Source: BankScope (Bureau Van Dijk)

r = herbank interest rate (interest rate Cycluses *i D*) *N*: Source: DanaScope (Dureau Van Dyk) *r* = Interbank interest rate %. Source: Bank of Spain
GDPGR= Growth of GDP %. Source: National Accounts (OECD)
GDP/POP=Per capita GDP (Thousand of constant US \$). Source: National Accounts (OECD)
BR/POP= Branches per population (Branches per thusand people). Source: Bank Profitability and National Accounts (OECD)

POP/Km²=Population / Extension (Thousand people / Km2). Source: National Accounts (OECD) and New Cronoss (Eurostat)

HHIA= Herfindahl index in total assets (calculated at national level). Source BankScope (Bureau Van Dijk) HHIL= Herfindahl index in the loan market (calculated at national level). Source BankScope (Bureau Van Dijk)

HHID= Herfindahl index in the deposit market (calculated at national level). Source BankScope (Bureau Van Dijk) LA= Log of average total assets (in thousand of US \$) in sample. Source: BankScope (Bureau Van Dijk)

BANKS= Number of banks in sample

Table 2. Interest rates and marginal costs

a) Loans

	Interest rat	$\operatorname{res}(r_L)$	marginal cos	sts (mc_L)	Absolute margin	$s(r_L - r - mc_L)$
	1993	2002	1993	2002	1993	2002
AUSTRIA	0.092	0.066	0.011	0.013	0.013	0.020
BELGIUM	0.096	0.061	0.009	0.010	0.005	0.018
DENMARK	0.125	0.061	0.015	0.011	0.009	0.017
FINLAND	0.099	0.064	0.016	0.011	0.008	0.020
FRANCE	0.096	0.066	0.011	0.011	0.002	0.023
GERMANY	0.095	0.065	0.011	0.010	0.012	0.023
GREECE	0.147	0.077	0.013	0.014	-0.096	0.031
IRELAND	0.129	0.062	0.015	0.005	0.028	0.024
ITALY	0.131	0.078	0.018	0.019	0.012	0.026
LUXEMBOURG	0.096	0.081	0.006	0.006	0.009	0.041
NETHERLANDS	0.088	0.069	0.014	0.010	0.006	0.026
PORTUGAL	0.165	0.084	0.016	0.013	0.023	0.039
SPAIN	0.142	0.075	0.015	0.012	0.010	0.030
SWEDEN	0.132	0.069	0.017	0.013	0.030	0.015
UNITED KINGDOM	0.098	0.051	0.015	0.011	0.026	0.000
UE-15	0.105	0.067	0.013	0.012	0.007	0.021

b) Deposits

	Interest rat	$es(r_D)$	marginal cos	sts (mc_D)	Absolute margin	s (<i>r-r_D -mc_D</i>)
	1993	2002	1993	2002	1993	2002
AUSTRIA	0.070	0.034	0.006	0.006	-0.008	-0.008
BELGIUM	0.074	0.036	0.007	0.008	0.001	-0.010
DENMARK	0.068	0.032	0.011	0.006	0.022	-0.005
FINLAND	0.071	0.026	0.009	0.006	-0.005	-0.001
FRANCE	0.072	0.054	0.008	0.007	0.003	-0.028
GERMANY	0.072	0.047	0.006	0.005	-0.007	-0.018
GREECE	0.122	0.029	0.010	0.011	0.096	-0.007
IRELAND	0.064	0.031	0.010	0.003	0.012	-0.001
ITALY	0.079	0.036	0.014	0.008	0.008	-0.011
LUXEMBOURG	0.084	0.084	0.003	0.003	-0.007	-0.053
NETHERLANDS	0.060	0.059	0.008	0.002	-0.001	-0.029
PORTUGAL	0.091	0.048	0.014	0.008	0.022	-0.023
SPAIN	0.080	0.029	0.013	0.009	0.024	-0.004
SWEDEN	0.084	0.035	0.008	0.006	-0.008	0.000
UNITED KINGDOM	0.057	0.026	0.012	0.006	-0.012	0.007
UE-15	0.073	0.043	0.009	0.006	0.001	-0.016

Source: BankScope and own elaboration

Table 3. Lerner index of market power

a) Loans

	1993	1994	1995	1996	1997	1998	1999	2000	2001	2002
AUSTRIA	0.140	0.271	0.334	0.416	0.383	0.327	0.351	0.182	0.230	0.307
BELGIUM	0.056	0.228	0.307	0.420	0.351	0.329	0.409	0.245	0.240	0.298
DENMARK	0.073	0.321	0.260	0.384	0.374	0.334	0.360	0.201	0.234	0.275
FINLAND	0.078	0.187	0.052	0.309	0.406	0.400	0.534	0.372	0.264	0.319
FRANCE	0.022	0.188	0.158	0.383	0.429	0.433	0.441	0.241	0.257	0.340
GERMANY	0.125	0.300	0.351	0.458	0.427	0.379	0.424	0.247	0.252	0.347
GREECE	-0.654	-0.479	-0.282	-0.153	-0.076	-0.156	0.013	0.144	0.379	0.397
IRELAND	0.221	0.280	0.277	0.196	0.159	0.185	0.360	0.307	0.328	0.388
ITALY	0.092	0.121	0.048	0.104	0.145	0.259	0.373	0.267	0.349	0.335
LUXEMBOURG	0.095	0.228	0.347	0.505	0.493	0.501	0.550	0.436	0.406	0.513
NETHERLANDS	0.072	0.245	0.342	0.445	0.438	0.444	0.411	0.284	0.293	0.376
PORTUGAL	0.139	-0.060	0.053	0.135	0.233	0.355	0.429	0.339	0.408	0.461
SPAIN	0.071	0.245	0.121	0.241	0.317	0.363	0.432	0.279	0.374	0.398
SWEDEN	0.227	0.190	0.154	0.284	0.300	0.332	0.370	0.278	0.217	0.213
UNITED KINGDOM	0.266	0.270	0.094	0.047	-0.151	-0.048	0.046	-0.039	0.017	0.006
UE-15	0.091	0.221	0.203	0.315	0.325	0.327	0.382	0.233	0.270	0.326

b) Deposits

	1993	1994	1995	1996	1997	1998	1999	2000	2001	2002
AUSTRIA	-0.115	-0.296	-0.395	-0.510	-0.462	-0.392	-0.436	-0.189	-0.160	-0.226
BELGIUM	0.010	-0.188	-0.349	-0.483	-0.386	-0.376	-0.463	-0.273	-0.254	-0.291
DENMARK	0.322	0.010	0.026	-0.252	-0.305	-0.286	-0.378	-0.225	-0.146	-0.154
FINLAND	-0.070	-0.125	0.015	-0.338	-0.366	-0.354	-0.625	-0.176	0.418	-0.019
FRANCE	0.040	-0.158	-0.149	-0.468	-0.554	-0.566	-0.601	-0.426	-0.415	-0.514
GERMANY	-0.094	-0.289	-0.383	-0.538	-0.524	-0.436	-0.515	-0.308	-0.314	-0.395
GREECE	0.788	0.621	0.476	0.326	0.298	0.403	0.171	-0.003	-0.192	-0.241
IRELAND	0.187	0.209	0.079	0.054	0.121	0.187	0.055	0.042	-0.105	-0.029
ITALY	0.107	0.030	0.185	-0.010	-0.100	-0.262	-0.441	-0.210	-0.306	-0.296
LUXEMBOURG	-0.078	-0.230	-0.400	-0.569	-0.586	-0.620	-0.667	-0.563	-0.521	-0.637
NETHERLANDS	-0.009	-0.233	-0.377	-0.572	-0.522	-0.530	-0.531	-0.415	-0.428	-0.485
PORTUGAL	0.240	0.329	0.225	0.050	-0.101	-0.261	-0.403	-0.266	-0.365	-0.475
SPAIN	0.302	0.088	0.221	0.002	-0.091	-0.168	-0.337	-0.054	-0.123	-0.154
SWEDEN	-0.091	-0.023	0.011	-0.217	-0.233	-0.336	-0.466	-0.352	-0.196	-0.006
UNITED KINGDOM	-0.208	-0.222	0.073	0.221	0.456	0.288	0.199	0.250	0.139	0.281
UE-15	0.014	-0.155	-0.149	-0.327	-0.375	-0.365	-0.460	-0.287	-0.310	-0.369

Source: BankScope and own elaboration

	1003	1004	1005	1007	1007	1000	1000	2000	2001	2002
	1993	1994	1995	1996	1997	1998	1999	2000	2001	2002
AUSTRIA	0.36	0.43	0.46	0.51	0.52	0.38	0.34	0.33	0.69	0.89
BELGIUM	0.27	0.31	0.06	0.13	0.15	0.07	0.18	0.09	0.11	0.28
DENMARK	2.12	2.04	1.54	1.41	1.11	0.92	0.68	0.82	1.13	1.14
FINLAND	0.12	0.27	0.12	0.18	0.35	0.44	0.44	0.57	0.72	0.61
FRANCE	0.18	0.36	0.25	0.35	0.27	0.19	0.20	-0.15	-0.05	0.03
GERMANY	0.41	0.78	0.73	0.80	0.54	0.67	0.60	0.29	0.29	0.57
GREECE	1.96	1.41	0.87	0.66	0.71	0.93	0.59	0.40	0.65	0.70
IRELAND	0.85	0.92	1.08	0.64	1.00	1.25	1.54	1.81	1.64	1.78
ITALY	1.05	0.79	0.88	0.48	0.34	0.47	0.66	0.66	0.92	0.79
LUXEMBOURG	-1.73	-5.66	-12.85	-15.21	-19.02	-18.86	-12.59	-14.36	-9.96	-11.39
NETHERLANDS	0.34	0.50	0.64	0.63	0.73	0.88	0.79	0.15	0.08	0.31
PORTUGAL	1.47	0.88	0.92	0.59	0.54	0.74	0.88	1.12	1.52	1.58
SPAIN	1.24	1.26	1.13	1.06	1.10	1.12	1.03	1.00	1.47	1.39
SWEDEN	1.13	0.80	0.73	0.94	0.93	0.91	0.84	0.57	0.57	0.77
UNITED KINGDOM	0.73	0.66	0.57	0.68	0.30	0.50	0.62	0.41	0.36	0.42
UE-15	0.54	0.62	0.53	0.54	0.42	0.49	0.50	0.27	0.42	0.54

Table 4. Welfare loss associated to market powerPercentage of GDP

The table shows the welfare loss associated with market power calculated according to expression (4) Source: BankScope, European Central Bank and own elaboration

Table 5. Mean efficiency scores

	1993	1994	1995	1996	1997	1998	1999	2000	2001	2002
AUSTRIA	0.870	0.868	0.867	0.866	0.864	0.867	0.880	0.832	0.822	0.821
BELGIUM	0.894	0.903	0.896	0.875	0.848	0.798	0.824	0.800	0.784	0.776
DENMARK	0.897	0.905	0.883	0.853	0.833	0.836	0.831	0.802	0.886	0.885
FINLAND	0.724	0.780	0.849	0.870	0.795	0.818	0.827	0.888	0.872	0.932
FRANCE	0.844	0.841	0.851	0.857	0.872	0.857	0.850	0.840	0.842	0.848
GERMANY	0.871	0.865	0.866	0.864	0.854	0.853	0.835	0.834	0.829	0.833
GREECE	0.905	0.910	0.909	0.899	0.891	0.906	0.874	0.878	0.874	0.880
IRELAND	-	-	0.897	0.903	0.889	0.913	0.761	0.852	0.845	0.919
ITALY	0.893	0.889	0.895	0.894	0.893	0.894	0.876	0.874	0.862	0.861
LUXEMBOURG	0.833	0.822	0.825	0.818	0.784	0.769	0.806	0.791	0.772	0.805
NETHERLANDS	0.742	0.747	0.749	0.738	0.770	0.774	0.769	0.796	-	-
PORTUGAL	0.828	0.836	0.881	0.854	0.852	0.848	0.838	0.825	0.766	0.740
SPAIN	0.864	0.872	0.881	0.884	0.886	0.884	0.877	0.873	0.872	0.881
SWEDEN	0.796	0.802	0.852	0.856	0.866	0.862	0.896	0.902	0.908	0.899
UNITED KINGDOM	0.838	0.835	0.843	0.853	0.854	0.838	0.875	0.887	0.907	0.899
UF-15	0.861	0.859	0.866	0.867	0.865	0.856	0.853	0.849	0.843	0 848

In Ireland (1993 and 1994) and the Netherlands (2002 and 2003) the efficiency scores have not been computed because the number of branches was not available

Source: BankScope and own elaboration

Table 6.	Welfare	loss associa	ated to ba	anks' co	st inefficiency
Percentage	of GDP				

	1993	1994	1995	1996	1997	1998	1999	2000	2001	2002
AUSTRIA	0.292	0.334	0.332	0.370	0.361	0.457	0.304	0.311	0.288	0.312
BELGIUM	0.318	0.309	0.310	0.425	0.709	1.444	0.778	0.995	0.979	1.107
DENMARK	0.200	0.193	0.287	0.341	0.371	0.407	0.389	0.497	0.259	0.284
FINLAND	0.319	0.349	0.234	0.281	0.741	0.702	0.555	0.323	0.079	0.109
FRANCE	0.512	0.561	0.530	0.512	0.476	0.561	0.590	0.697	0.648	0.587
GERMANY	0.300	0.371	0.380	0.386	0.448	0.479	0.532	0.495	0.529	0.473
GREECE	0.157	0.166	0.174	0.214	0.245	0.213	0.253	0.284	0.283	0.288
IRELAND	-	-	0.035	0.040	0.037	0.024	0.041	0.013	0.014	0.007
ITALY	0.308	0.368	0.349	0.361	0.359	0.352	0.358	0.409	0.458	0.471
LUXEMBOURG	1.697	2.447	2.330	2.002	2.533	3.367	2.215	2.837	2.732	2.457
NETHERLANDS	0.049	0.057	0.073	0.103	0.065	0.074	0.069	0.060	-	-
PORTUGAL	0.485	0.570	0.435	0.567	0.596	0.561	0.545	0.597	0.597	0.613
SPAIN	0.444	0.451	0.411	0.377	0.372	0.397	0.356	0.386	0.379	0.384
SWEDEN	0.265	0.290	0.271	0.271	0.255	0.276	0.186	0.187	0.184	0.160
UNITED KINGDOM	0.335	0.348	0.329	0.327	0.251	0.302	0.235	0.157	0.108	0.126
UE-15	0.345	0.388	0.372	0.378	0.388	0.444	0.417	0.423	0.422	0.402

Source: BankScope and own elaboration

	(1)	(2)	(3)
Lerner (loans)			0.0729 ***
			(0.01)
Lerner (deposits)			0.0301 ***
			(0.01)
HHIA (Total assets)	-0.1108		
	(0.29)		
HHIL (loans)		2.2586 ***	
		(0.35)	
HHID (deposits)		-1.8071 ***	
		(0.34)	
Size (log of total assets)	0.0842 ***	0.0905 ***	0.0862 ***
	(0.01)	(0.01)	(0.01)
S1(commercial banks)	0.0699	0.0596	0.0657
	(0.20)	(0.20)	(0.20)
S2(savings banks)	0.0734	0.0669	0.0681
	(0.22)	(0.22)	(0.22)
S3 (cooperative banks)	0.1037	0.0939	0.0996
	(0.21)	(0.21)	(0.21)
Adjusted R2	0.6669	0.6675	0.6674
Hausman Test (p-value)	52.6000	79.8000	88.7700
	(0.00)	(0.00)	(0.00)
Obervations	29,694	29,694	26,964
Elasticities of effi	ciency to changes in	the independent va	riables
Percentage change in the	efficiency due to a 100%	change in each independe	nt variable
Lerner (loans)			0.2938
Lerner (deposits)			0.0461
HHIA	-0.0489		
HHIL (loans)		0.9925	
HHID (deposits)		-0.8611	
Size (total assets)	1.8080	1.8589	1.8255

Table 7. Determinants of cost efficiency

All models were estimated using fixed effects and time effects (the Hausman test suggests that a fixed effects model is more appropriate). As efficiency is a variable bounded between 0 and 1, the log of a logistic transformation, $\ln [EF / (1-EF)]$, is used to estimate the determinant of cost efficiency. The excluded dummy specialization is S4 (Other type of banks). In brackets standard deviations

The elasticities with respect to total assets have been calculated taking into account that in the estimation the variable is in log. So, the elasticity values shown reflect the change in efficiency due to the change in total assets of the bank

*** Significant at 1%, ** Significant at 5% and * significant at 10%. Source: BankScope (Bureau Van Dijk) and own elaboration.

	1993	1994	1995	1996	1997	1998	1999	2000	2001	2002
AUSTRIA	0.25	0.23	0.61	0.42	0.25	0.00	0.05	0.21	0.32	0.24
BELGIUM	0.00	0.03	0.35	0.33	0.08	0.01	0.09	0.08	0.16	0.19
DENMARK	1.42	1.09	0.83	0.56	0.46	0.30	0.25	0.08	0.18	0.18
FINLAND	0.00	0.04	0.22	0.00	0.15	0.16	0.28	0.36	0.37	0.23
FRANCE	0.06	0.00	0.00	0.22	0.35	0.35	0.18	0.03	0.00	0.04
GERMANY	0.20	0.43	0.77	0.72	0.51	0.26	0.20	0.07	0.00	0.10
GREECE	2.95	2.85	1.76	1.91	1.96	2.54	2.04	0.90	0.54	0.36
IRELAND	0.57	0.49	0.60	0.56	1.12	1.21	0.75	1.14	0.86	0.88
ITALY	0.37	0.58	1.08	0.93	0.53	0.16	0.02	0.21	0.31	0.11
LUXEMBOURG	0.10	0.17	3.70	3.89	3.62	4.56	4.28	5.57	3.75	4.44
NETHERLANDS	0.00	0.10	0.62	0.57	0.57	0.66	0.16	0.30	0.14	0.28
PORTUGAL	1.03	1.57	1.24	1.15	0.77	0.36	0.31	0.64	0.87	0.81
SPAIN	0.86	0.68	1.01	0.87	0.59	0.47	0.35	0.62	0.87	0.63
SWEDEN	0.78	0.19	0.27	0.17	0.16	0.05	0.00	0.15	0.11	0.30
UNITED KINGDOM	0.83	0.21	0.51	1.17	1.75	1.62	1.37	1.23	0.94	0.92
UE-15	0.31	0.28	0.53	0.63	0.67	0.52	0.36	0.33	0.30	0.32

Table 8. Social welfare gains associated to a reduction in market powerPercentage of GDP

Table 8 shows the social welfare gains associated with market power if countries with Lerner Indices in loans and deposits above the EU level were to converge to the average EU level, calculated according to expression (4)

Source: BankScope, European Central Bank and own elaboration

	1993	1994	1995	1996	1997	1998	1999	2000	2001	2002
AUSTRIA	0.0009	0.0010	0.0026	0.0022	0.0012	0.0000	0.0002	0.0006	0.0008	0.0008
BELGIUM	0.0000	0.0002	0.0025	0.0028	0.0010	0.0001	0.0009	0.0007	0.0008	0.0013
DENMARK	0.0020	0.0027	0.0025	0.0018	0.0014	0.0008	0.0006	0.0005	0.0012	0.0017
FINLAND	0.0000	0.0002	0.0008	0.0000	0.0024	0.0024	0.0039	0.0042	0.0015	0.0018
FRANCE	0.0003	0.0000	0.0000	0.0019	0.0031	0.0033	0.0019	0.0003	0.0000	0.0004
GERMANY	0.0006	0.0017	0.0033	0.0032	0.0025	0.0013	0.0011	0.0003	0.0000	0.0005
GREECE	0.0042	0.0047	0.0039	0.0045	0.0049	0.0057	0.0041	0.0022	0.0028	0.0023
IRELAND	-	-	0.0005	0.0005	0.0005	0.0005	0.0003	0.0001	0.0001	0.0001
ITALY	0.0009	0.0020	0.0036	0.0035	0.0030	0.0011	0.0002	0.0017	0.0021	0.0011
LUXEMBOURG	0.0003	0.0008	0.0153	0.0165	0.0156	0.0201	0.0152	0.0218	0.0129	0.0187
NETHERLANDS	0.0000	0.0000	0.0003	0.0004	0.0003	0.0003	0.0001	0.0001	-	-
PORTUGAL	0.0032	0.0055	0.0045	0.0048	0.0036	0.0021	0.0019	0.0031	0.0028	0.0025
SPAIN	0.0031	0.0035	0.0042	0.0035	0.0030	0.0032	0.0023	0.0034	0.0042	0.0041
SWEDEN	0.0014	0.0006	0.0010	0.0007	0.0009	0.0003	0.0000	0.0007	0.0007	0.0019
UNITED KINGDOM	0.0029	0.0008	0.0015	0.0040	0.0046	0.0040	0.0041	0.0024	0.0017	0.0027
UE-15	0.0011	0.0013	0.0023	0.0029	0.0029	0.0022	0.0017	0.0013	0.0011	0.0014

 Table 9: Cost efficiency loss associated with a decrease in market power

 Percentage of GDP

Source: BankScope, European Central Bank, Eurostat and own elaboration



Figure 1. The Harberger's triangle

DOCUMENTOS DE TRABAJO

Últimos números publicados

159/2000	Participación privada en la construcción y explotación de carreteras de peaje Ginés de Rus, Manuel Romero y Lourdes Trujillo
160/2000	Errores y posibles soluciones en la aplicación del Value at Risk Mariano González Sánchez
161/2000	Tax neutrality on saving assets. The spahish case before and after the tax reform Cristina Ruza y de Paz-Curbera
162/2000	Private rates of return to human capital in Spain: new evidence F. Barceinas, J. Oliver-Alonso, J.L. Raymond y J.L. Roig-Sabaté
163/2000	El control interno del riesgo. Una propuesta de sistema de límites riesgo neutral Mariano González Sánchez
164/2001	La evolución de las políticas de gasto de las Administraciones Públicas en los años 90 Alfonso Utrilla de la Hoz y Carmen Pérez Esparrells
165/2001	Bank cost efficiency and output specification Emili Tortosa-Ausina
166/2001	Recent trends in Spanish income distribution: A robust picture of falling income inequality Josep Oliver-Alonso, Xavier Ramos y José Luis Raymond-Bara
167/2001	Efectos redistributivos y sobre el bienestar social del tratamiento de las cargas familiares en el nuevo IRPF Nuria Badenes Plá, Julio López Laborda, Jorge Onrubia Fernández
168/2001	The Effects of Bank Debt on Financial Structure of Small and Medium Firms in some Euro- pean Countries Mónica Melle-Hernández
169/2001	La política de cohesión de la UE ampliada: la perspectiva de España Ismael Sanz Labrador
170/2002	Riesgo de liquidez de Mercado Mariano González Sánchez
171/2002	Los costes de administración para el afiliado en los sistemas de pensiones basados en cuentas de capitalización individual: medida y comparación internacional. José Enrique Devesa Carpio, Rosa Rodríguez Barrera, Carlos Vidal Meliá
172/2002	La encuesta continua de presupuestos familiares (1985-1996): descripción, representatividad y propuestas de metodología para la explotación de la información de los ingresos y el gasto. Llorenc Pou, Joaquín Alegre
173/2002	Modelos paramétricos y no paramétricos en problemas de concesión de tarjetas de credito. Rosa Puertas, María Bonilla, Ignacio Olmeda

174/2002	Mercado único, comercio intra-industrial y costes de ajuste en las manufacturas españolas. José Vicente Blanes Cristóbal
175/2003	La Administración tributaria en España. Un análisis de la gestión a través de los ingresos y de los gastos. Juan de Dios Jiménez Aguilera, Pedro Enrique Barrilao González
176/2003	The Falling Share of Cash Payments in Spain. Santiago Carbó Valverde, Rafael López del Paso, David B. Humphrey Publicado en "Moneda y Crédito" nº 217, pags. 167-189.
177/2003	Effects of ATMs and Electronic Payments on Banking Costs: The Spanish Case. Santiago Carbó Valverde, Rafael López del Paso, David B. Humphrey
178/2003	Factors explaining the interest margin in the banking sectors of the European Union. Joaquín Maudos y Juan Fernández Guevara
179/2003	Los planes de stock options para directivos y consejeros y su valoración por el mercado de valores en España. Mónica Melle Hernández
180/2003	Ownership and Performance in Europe and US Banking – A comparison of Commercial, Co- operative & Savings Banks. Yener Altunbas, Santiago Carbó y Phil Molyneux
181/2003	The Euro effect on the integration of the European stock markets. Mónica Melle Hernández
182/2004	In search of complementarity in the innovation strategy: international R&D and external knowledge acquisition. Bruno Cassiman, Reinhilde Veugelers
183/2004	Fijación de precios en el sector público: una aplicación para el servicio municipal de sumi- nistro de agua. Mª Ángeles García Valiñas
184/2004	Estimación de la economía sumergida es España: un modelo estructural de variables latentes. Ángel Alañón Pardo, Miguel Gómez de Antonio
185/2004	Causas políticas y consecuencias sociales de la corrupción. Joan Oriol Prats Cabrera
186/2004	Loan bankers' decisions and sensitivity to the audit report using the belief revision model. Andrés Guiral Contreras and José A. Gonzalo Angulo
187/2004	El modelo de Black, Derman y Toy en la práctica. Aplicación al mercado español. Marta Tolentino García-Abadillo y Antonio Díaz Pérez
188/2004	Does market competition make banks perform well?. Mónica Melle
189/2004	Efficiency differences among banks: external, technical, internal, and managerial Santiago Carbó Valverde, David B. Humphrey y Rafael López del Paso

190/2004	Una aproximación al análisis de los costes de la esquizofrenia en españa: los modelos jerár- quicos bayesianos F. J. Vázquez-Polo, M. A. Negrín, J. M. Cavasés, E. Sánchez y grupo RIRAG
191/2004	Environmental proactivity and business performance: an empirical analysis Javier González-Benito y Óscar González-Benito
192/2004	Economic risk to beneficiaries in notional defined contribution accounts (NDCs) Carlos Vidal-Meliá, Inmaculada Domínguez-Fabian y José Enrique Devesa-Carpio
193/2004	Sources of efficiency gains in port reform: non parametric malmquist decomposition tfp in- dex for Mexico Antonio Estache, Beatriz Tovar de la Fé y Lourdes Trujillo
194/2004	Persistencia de resultados en los fondos de inversión españoles Alfredo Ciriaco Fernández y Rafael Santamaría Aquilué
195/2005	El modelo de revisión de creencias como aproximación psicológica a la formación del juicio del auditor sobre la gestión continuada Andrés Guiral Contreras y Francisco Esteso Sánchez
196/2005	La nueva financiación sanitaria en España: descentralización y prospectiva David Cantarero Prieto
197/2005	A cointegration analysis of the Long-Run supply response of Spanish agriculture to the com- mon agricultural policy José A. Mendez, Ricardo Mora y Carlos San Juan
198/2005	¿Refleja la estructura temporal de los tipos de interés del mercado español preferencia por la li- quidez? Magdalena Massot Perelló y Juan M. Nave
199/2005	Análisis de impacto de los Fondos Estructurales Europeos recibidos por una economía regional: Un enfoque a través de Matrices de Contabilidad Social M. Carmen Lima y M. Alejandro Cardenete
200/2005	Does the development of non-cash payments affect monetary policy transmission? Santiago Carbó Valverde y Rafael López del Paso
201/2005	Firm and time varying technical and allocative efficiency: an application for port cargo han- dling firms Ana Rodríguez-Álvarez, Beatriz Tovar de la Fe y Lourdes Trujillo
202/2005	Contractual complexity in strategic alliances Jeffrey J. Reuer y Africa Ariño
203/2005	Factores determinantes de la evolución del empleo en las empresas adquiridas por opa Nuria Alcalde Fradejas y Inés Pérez-Soba Aguilar
204/2005	Nonlinear Forecasting in Economics: a comparison between Comprehension Approach versus Learning Approach. An Application to Spanish Time Series Elena Olmedo, Juan M. Valderas, Ricardo Gimeno and Lorenzo Escot

205/2005	Precio de la tierra con presión urbana: un modelo para España Esther Decimavilla, Carlos San Juan y Stefan Sperlich
206/2005	Interregional migration in Spain: a semiparametric analysis Adolfo Maza y José Villaverde
207/2005	Productivity growth in European banking Carmen Murillo-Melchor, José Manuel Pastor y Emili Tortosa-Ausina
208/2005	Explaining Bank Cost Efficiency in Europe: Environmental and Productivity Influences. Santiago Carbó Valverde, David B. Humphrey y Rafael López del Paso
209/2005	La elasticidad de sustitución intertemporal con preferencias no separables intratemporalmente: los casos de Alemania, España y Francia. Elena Márquez de la Cruz, Ana R. Martínez Cañete y Inés Pérez-Soba Aguilar
210/2005	Contribución de los efectos tamaño, book-to-market y momentum a la valoración de activos: el caso español. Begoña Font-Belaire y Alfredo Juan Grau-Grau
211/2005	Permanent income, convergence and inequality among countries José M. Pastor and Lorenzo Serrano
212/2005	The Latin Model of Welfare: Do 'Insertion Contracts' Reduce Long-Term Dependence? Luis Ayala and Magdalena Rodríguez
213/2005	The effect of geographic expansion on the productivity of Spanish savings banks Manuel Illueca, José M. Pastor and Emili Tortosa-Ausina
214/2005	Dynamic network interconnection under consumer switching costs Ángel Luis López Rodríguez
215/2005	La influencia del entorno socioeconómico en la realización de estudios universitarios: una aproximación al caso español en la década de los noventa Marta Rahona López
216/2005	The valuation of spanish ipos: efficiency analysis Susana Álvarez Otero
217/2005	On the generation of a regular multi-input multi-output technology using parametric output dis- tance functions Sergio Perelman and Daniel Santin
218/2005	La gobernanza de los procesos parlamentarios: la organización industrial del congreso de los di- putados en España Gonzalo Caballero Miguez
219/2005	Determinants of bank market structure: Efficiency and political economy variables Francisco González
220/2005	Agresividad de las órdenes introducidas en el mercado español: estrategias, determinantes y me- didas de performance David Abad Díaz

221/2005	Tendencia post-anuncio de resultados contables: evidencia para el mercado español Carlos Forner Rodríguez, Joaquín Marhuenda Fructuoso y Sonia Sanabria García
222/2005	Human capital accumulation and geography: empirical evidence in the European Union Jesús López-Rodríguez, J. Andrés Faíña y Jose Lopez Rodríguez
223/2005	Auditors' Forecasting in Going Concern Decisions: Framing, Confidence and Information Proc- essing Waymond Rodgers and Andrés Guiral
224/2005	The effect of Structural Fund spending on the Galician region: an assessment of the 1994-1999 and 2000-2006 Galician CSFs José Ramón Cancelo de la Torre, J. Andrés Faíña and Jesús López-Rodríguez
225/2005	The effects of ownership structure and board composition on the audit committee activity: Span- ish evidence Carlos Fernández Méndez and Rubén Arrondo García
226/2005	Cross-country determinants of bank income smoothing by managing loan loss provisions Ana Rosa Fonseca and Francisco González
227/2005	Incumplimiento fiscal en el irpf (1993-2000): un análisis de sus factores determinantes Alejandro Estellér Moré
228/2005	Region versus Industry effects: volatility transmission Pilar Soriano Felipe and Francisco J. Climent Diranzo
229/2005	Concurrent Engineering: The Moderating Effect Of Uncertainty On New Product Development Success Daniel Vázquez-Bustelo and Sandra Valle
230/2005	On zero lower bound traps: a framework for the analysis of monetary policy in the 'age' of cen- tral banks Alfonso Palacio-Vera
231/2005	Reconciling Sustainability and Discounting in Cost Benefit Analysis: a methodological proposal M. Carmen Almansa Sáez and Javier Calatrava Requena
232/2005	Can The Excess Of Liquidity Affect The Effectiveness Of The European Monetary Policy? Santiago Carbó Valverde and Rafael López del Paso
233/2005	Inheritance Taxes In The Eu Fiscal Systems: The Present Situation And Future Perspectives. Miguel Angel Barberán Lahuerta
234/2006	Bank Ownership And Informativeness Of Earnings. Víctor M. González
235/2006	Developing A Predictive Method: A Comparative Study Of The Partial Least Squares Vs Maxi- mum Likelihood Techniques. Waymond Rodgers, Paul Pavlou and Andres Guiral.
236/2006	Using Compromise Programming for Macroeconomic Policy Making in a General Equilibrium Framework: Theory and Application to the Spanish Economy. Francisco J. André, M. Alejandro Cardenete y Carlos Romero.

237/2006	Bank Market Power And Sme Financing Constraints. Santiago Carbó-Valverde, Francisco Rodríguez-Fernández y Gregory F. Udell.
238/2006	Trade Effects Of Monetary Agreements: Evidence For Oecd Countries. Salvador Gil-Pareja, Rafael Llorca-Vivero y José Antonio Martínez-Serrano.
239/2006	The Quality Of Institutions: A Genetic Programming Approach. Marcos Álvarez-Díaz y Gonzalo Caballero Miguez.
240/2006	La interacción entre el éxito competitivo y las condiciones del mercado doméstico como deter- minantes de la decisión de exportación en las Pymes. Francisco García Pérez.
241/2006	Una estimación de la depreciación del capital humano por sectores, por ocupación y en el tiempo. Inés P. Murillo.
242/2006	Consumption And Leisure Externalities, Economic Growth And Equilibrium Efficiency. Manuel A. Gómez.
243/2006	Measuring efficiency in education: an analysis of different approaches for incorporating non-discretionary inputs. Jose Manuel Cordero-Ferrera, Francisco Pedraja-Chaparro y Javier Salinas-Jiménez
244/2006	Did The European Exchange-Rate Mechanism Contribute To The Integration Of Peripheral Countries?. Salvador Gil-Pareja, Rafael Llorca-Vivero y José Antonio Martínez-Serrano
245/2006	Intergenerational Health Mobility: An Empirical Approach Based On The Echp. Marta Pascual and David Cantarero
246/2006	Measurement and analysis of the Spanish Stock Exchange using the Lyapunov exponent with digital technology. Salvador Rojí Ferrari and Ana Gonzalez Marcos
247/2006	Testing For Structural Breaks In Variance Withadditive Outliers And Measurement Errors. Paulo M.M. Rodrigues and Antonio Rubia
248/2006	The Cost Of Market Power In Banking: Social Welfare Loss Vs. Cost Inefficiency. Joaquín Maudos and Juan Fernández de Guevara